Author's response to reviews

Title: Cost, affordability and cost-effectiveness of strategies to control tuberculosis in countries with high HIV prevalence

Authors:

Christine SM Currie (christine.currie@soton.ac.uk)
Katherine Floyd (floydk@who.int)
Brian G Williams (williamsbg@who.int)
Christopher Dye (dyec@who.int)

Version: 7 Date: 7 December 2005

Author's response to reviews:

As requested, the subdivision of Tables 3-5 has been removed. Apologies, I did not understand what was required before.